Author's response to reviews

Title: Extending the spectrum of Ellis van Creveld syndrome: a large family with a mild mutation in the EVC gene

Authors:

Hakan Ulucan (ulucanh@mail.nih.gov)
Davut Gül (davutgul2001@yahoo.com)
Julie C Sapp (sappj@mail.nih.gov)
John Cockerham (jcockerh@cnmc.org)
Jennifer J Johnston (jjohnsto@mail.nih.gov)
Leslie G Biesecker (leslieb@helix.nih.gov)

Version: 2 Date: 19 June 2008

Author's response to reviews:

The consents for publishing clinical information have been obtained from patients.

The text below is included in the Acknowledgements sections as asked from the editor before submission:

"Written consent was obtained from the patient or their relative for publication of the patient's details."